Refocusing Agricultural Subsidy for Food Security in Nigeria

Sulaiman A. Yusuf*

I. Introduction

ood security has remained a key human right-related issue and ranks second only to poverty on the Sustainable Development Goals (SDGs). The third SDG of good health and well-being is intricately linked to the first two SDGs. At the core of these, is the physical, social, and economic access to sufficient, safe, and nutritious food to meet dietary needs and food preferences for an active and healthy life for all people at all times.

Shaw (2007) defined food security as a situation 'when all people, at all times, have physical, social and economic access to sufficient, safe and nutritious food to meet their dietary needs and food preferences for an active and healthy life'. At the World Summit on Food Security in 2009, supported by World Health Organisation (WHO) and Food and Agricultural Organisation (FAO), this definition was reaffirmed, and the concept was extended and specified by adding the four pillars of food security which include food availability, access, utilisation and stability (Ecker & Breisinger, 2012). Table 1 summarises the different dimensions of food security and their definitions.

Dimensions of Food Security	Means of Identification
Physical AVAILABILITY of food	Food availability addresses the "supply side" of food security and is determined by the level of food production, stock levels and net trade.
Economic and physical ACCESS to food	An adequate supply of food at the national or international level does not in itself guarantee household-level food security. Concerns about insufficient food access have resulted in a greater policy focus on incomes, expenditure, markets, and prices in achieving food security objectives.
Food UTILISATION	Utilisation is commonly understood as the way the body makes the most of various nutrients in the food. Sufficient energy and nutrient intake by individuals are the result of good care and feeding practices, food preparation, diversity of the diet and

Table 1: Dimensions o	of Food Security
-----------------------	------------------

^{*} Prof. Sulaiman A. Yusuf is the Head, Department of Agricultural Economics, University of Ibadan, Ibadan. The support of Ayoola Adewale, Muhammed Adesina, James Fasakin, Haneefah Lawal and Paul Adeosun (all Doctoral Students) in the preparation of this paper is highly acknowledged. The usual disclaimer applies.

	intra-household distribution of food. Combined with good biological utilisation of food consumed, this determines the nutritional status of individuals.
STABILITY of the other three dimensions over time	Even if your food intake is adequate today, you are still considered to be food insecure if you have inadequate access to food on a periodic basis, risking a deterioration of your nutritional status. Adverse weather conditions, political instability, or economic factors (unemployment, rising food prices) may have an impact on your food security status.

Source: Adapted from Food and Agriculture Organisation, (2008).

It must be noted that food security objectives are only achievable when all the four dimensions of food security are simultaneously achieved.

Agriculture is at the centre of food security. In ensuring agricultural productivity and enhancing output, agricultural subsidy is a major tool. Agricultural subsidies can be in form of cash payments to farmers to support their operations or indirectly through the reduction of prices of major agricultural inputs to address market failures and promote the adoption of new technologies (Fan et al., 2008). The idea was originally instituted to stabilise markets, help low-income farmers, and aid rural development.

Government agricultural subsidies are categorised into three groups: a production subsidy; an investment subsidy and an entry subsidy. A production subsidy tries to offset the production cost for farmers in a bid to boost output and mitigate losses. The objective of production subsidies is to expand production of a particular product more so that the market would promote it without raising the final price to consumers. We have direct and indirect subsidies. Direct subsidy occurs when government transfers cash straight to the beneficiaries. On the other hand, the indirect subsidy is a situation of no specific cash (monetary amount) paid to the beneficiaries. Rather, it comes in form of government-backed loans or payments in kind.

Figures 1 and 2 show the subsidy situation for some OECD countries. Figure 1 shows that South Africa provides the lowest percentage of agricultural support relative to gross farm receipts, while Israel provides the most agricultural support per gross farm receipts.





Source-OECD, 2021.



Figure 2: Selected Countries where Agriculture as a Share of GDP is most Subsidised (2019)

Source: Statista (2021).

Figure 2 shows that the Philippines has the highest agricultural support at 3.1 per cent in 2019. The USA committed 0.5 per cent of its GDP to agricultural support. This implies for every dollar of the GDP half a cent is committed to agriculture in the USA.

In the light of the above the paper asks two questions. Is the subsidy regime in the agricultural sector sufficiently situated to support food security? and is there a need to refocus the regime for higher productivity in the sector and food security? To answer these questions the paper reviews the current subsidy regime in Nigeria and in other climes to draw relevant lessons. Thus, the objectives of this paper are to identify the gaps in agriculture subsidy regime; and to identify options for refocusing of the subsidy regime to improve food security.

Following the introduction, section 2 explores the nexus of agricultural subsidy and food security, while section 3 reviews the agricultural subsidy regime in Nigeria. Section 4 discusses the input-output dimensions of agricultural subsidy in Nigeria, and in section 5, the paper takes a dive into the agricultural subsidies from international perspective. In section 6, it proposes a paradigm shift for agricultural subsidy for food security, while section 7 concludes the paper.

II. Nexus of Agricultural Subsidy and Food Security

II.1 How will/does Agricultural Subsidy implementation translate to Food Security?

Food production subsidy is a broadly used tool to achieve food security in many developed countries, through the adoption of protectionist policies not only in the agro-industry, but also in others. Subsidy polices very often are used to overcome certain issues such as supporting uncompetitive industries or provide a buffer to an infant industry to influence market participants and therefore to adjust their market behaviour. Although the debate on implications of the continued use of subsidies in supporting food security remains unsettled, the standard subsidy theory is that it helps to increase agricultural output supply, which leads to output price reduction. Figure 3 provides the graphical illustration of the effect of subsidy on output and price of a given agricultural commodity.



Figure 3: Agricultural Subsidy (with trade)

As depicted in the graph, subsidy to domestic producers will shift the domestic supply curve downwards indicating more output because of the subsidy. Prior to the introduction of the subsidy, domestic producers supplied quantity Q1 and they imported Q3-Q1 of the commodity. Given the subsidy and at the world price Pw, domestic producers will produce quantity Q2, while the country only imports Q3-Q2. As a result of the subsidy, the more efficient foreign producers lose at the expense of the less efficient domestic producers. Triangle X represents the deadweight loss to the society indicating overallocation of resources by domestic producers because of government intervention. Consumers still pay Pw without a loss of consumer surplus. However, because consumers' tax is used to subsidise inefficient domestic producers, they may be worse off after the subsidy. In simple terms, agriculture subsidies, to the extent that they provide incentives to producers to increase production, impact on agricultural trade balances, either by reducing the import requirements of, or by increasing levels of exports from, the subsidising country. This results in an increase in excess supply on global markets and, where there are no demand shifters and there is a substantial increase in excess supply from one or more subsidising countries, leading to reduced global prices. These reduced global prices have potential impacts on import and/or export prices for all countries, affecting domestic market prices and the returns to producers and expenditure of consumers. This in turn, impacts development related indicators.

Economic and Financial Review

December 2021

The case for agricultural subsidies in Nigeria or elsewhere is justified by Idachaba (1994) as cited in Idachaba (2014) to include: attainment of desired resource use patterns, market promotion in the early stages of agricultural transition, agricultural subsidies as income transfer, agricultural subsidies to support agriculture as an infant industry and agricultural subsidies as compensating distortions within the theory of second best. Agricultural subsidies in Nigeria can further be justified by the worsening poverty estimated at 40.0 per cent in 2019 which is more pronounced in rural areas at 52.1 per cent (NBS, 2020). The income inequality of 35.1 per cent is also a call for concern just as the raging insecurity in the country. Poverty is more pronounced among male and female agricultural households, compared with other employment categories in rural, urban areas in the country.

III. Agricultural Subsidy Regimes in Nigeria

III.1 Is there an Agricultural Subsidy Framework in Nigeria?

Agricultural development in Nigeria has transcended different stages. Starting from the post-independence period, and navigating through different decades, agricultural programmes especially those that focus on subsidy to the farmers have witnessed many challenges that might make one argue against continuous agricultural subsidy in Nigeria. Prior to the decade of the 1960s, the central role of agriculture in Nigeria's economy was taken for granted with very little support from government. Despite this, Nigerian agriculture was able to grow at a sufficient rate to provide adequate food for an increasing population, supplying raw materials for a budding industrial sector, increasing public revenue and foreign exchange for government and employment opportunities for an expanding labour force. There was report of success emanating from the little support provided by government for agricultural development, as the support was concentrated on export crops like cocoa, groundnut, palm produce, rubber and cotton since self-sufficiency in food production seemed not to pose any problem worthy of public attention then. This made the regional government specialise more on the commodities they had comparative advantages on, such that the western region was noted for cocoa production, the eastern region for oil palm and the northern region for groundnut.

The challenges to Nigerian agriculture started to surface from the first decade of the country's independence through the second decade of the oil boom, resulting in a worsening and rapid deterioration of the country's agricultural situation. To tackle these problems, government initiated several agricultural policies, programmes and projects, largely within the framework of three successive rational development plans from 1970 to 1974, 1975 to 1980 and 1981 to 1985. Experience from these policies, programmes and projects has, however, suggested that there is no alternative to well-designed and articulate agricultural policies as instruments for promoting agricultural growth and development in Nigeria. Accordingly, the government has adopted a comprehensive package of policy instruments to

further develop and improve the agriculture sector especially in subsidies distributions.

Input subsidies are grants given by the government to farmers to reduce their production cost and improve their profit margin. Over the years, the Nigerian government has been making considerable expenditure on the provision of subsidised farm inputs (especially fertiliser). According to Takeshima and Liverpool-Tasie (2013), fertiliser subsidy alone constituted nearly 68.0 per cent of government agricultural expenditure in the past. Specifically, the cost of subsidy per farmer was between N22,125 to N24,825 depending on the type of crop the farmer cultivates (Federal Ministry of Agriculture and Rural Development, 2016). This has overtime constituted a huge burden on the Federal budget, raising concerns about its sustainability considering the dwindling financial resources of the country (Ayoola & Ayoola, 2016). Subsidy programmes are sustainable if they can be maintained over the long term without draining the public resources, or if the outcomes in terms of wider adoption of agricultural inputs and improved agricultural productivity persist after their termination (Baltzer & Hansen, 2011; Druilhe & Barreiro-hurlé, 2012).

A meso-analysis of the Nigeria Growth Enhancement Support Scheme by Michael et al. (2018) of different publications on the scheme, reveals that Nigeria's past agricultural input subsidy policy (known as the Market Stabilisation Scheme) was widely considered ineffective. The reasons cited, among others include: massive diversion of supply to benefit the middlemen; elite capture; the cumbersome acquisition process; and late delivery of inputs, as well as low quality of some of the input.

The GESS which ensured that farm input like fertilizer, insecticides/ herbicides, and improved seeds are disbursed to registered individual smallholder farmers through an electronic system had as its basis register of 10.5 million farmers across all states of the federation and FCT. As many as 14.4 Million farmers were reached by the scheme between 2012-2014 delivering 1.8 million MT of fertilizer and 174,000 MT of improved seeds respectively delivered to farmers. These combined led to 20.1 Million MT of additional food produced (Oredipe, 2015).

Despite the success recorded in the GESS, it is difficult to conclude that there is a sustainable framework for subsidy regime in Nigeria. This axiom is justified on the fact that farmers in the country have been yearning for such a smart subsidy programme that was able to record a great success in the transformation of agricultural sector. In recent years, the Nigerian economy has faced recession which has affected the capacity of governments to fund programmes like the GESS. Nigeria government is still heavily indebted to input suppliers for inputs they supplied in the past. This has a serious implication on the effectiveness and sustainability of the Scheme for its

December 2021

planned duration since such debts must be settled and other funds will still be sourced for future implementation (FMARD, 2016). It must be stated that as at July 2021, the agricultural subsidy for the year put at an approved N12.30 billion targeted at 2.2 million farmers is still in the works (Olisah, 2019 curled from Nairametrics, July 19, 2021).

IV. Input/Output Dimensions of Agricultural Subsidy in Nigeria

IV.1 Seed and Fertilizer Subsidy

Fertilizer subsidy programme in Nigeria dates back to the 1970s. This programme has however witnessed instabilities and inconsistencies as one government succeeds another in the country (Salimonu, 2007). The expected gains have been transferred to unintended beneficiaries at the expense of government treasury. It became a bigger problem between 1990 and 1996 when fertiliser subsidy expenditure consistently exceeded total capital on agriculture (Eboh et al., 2006). Also, fertiliser prices had been fluctuating such that farmers could not afford the market prices thus calling for the reshaping of the subsidy programme (Salman, 2019).

In 2013, the Government provided subsidy of up to 85.0 per cent discount for a bag of fertiliser which could explain the rise in Agricultural GDP from ¥16,816.55 billion in 2013 to ¥18,018.61 billion in 2014 although the reinforcement in farming and other agricultural activity has been minimal since the discovery of oil. Though the subsidy on fertiliser was well intended, the outcome was lower than expected.

Table 2 presents the subsidy cost and value of subsidy, as well as the quantity of fertiliser and cost of fertiliser supplied between 2001 and 2015.

Year		Quantity of Fertilizer Supplied (Mt)	Cost of Quantity Supplied (Ħ)	Subsidy Cost (벆)	Value of Subsidy (Ħ / Mt)	Rate of Subsidy (per cent)
2001	~	164,012	4,876,554,998	1,683,000,000	10,261	35
2002	010	163,700	3,605,662,509	1,485,000,000	9,071	41
2003	tior -20	511,841	4,620,418,025	1,188,000,000	2,321	26
2004	isa [.] 001	560,150	11,024,019,200	2,459,160,000	4,390	22
2005	abil S, 2	600,000	8,341,772,360	1,750,432,213	2,917	21
2006	Stc MS:	09,000	16,258,649,932	3,507,200,000	4,946	22
2007	ket e (I	990,000	19,422,363,970	4,855,590,994	4,904	25
2008	Aar em	691,153	57,055,503,960	14,263,875,990	20,637	25
2009	<pre>v</pre>	371,062	38,050,847,750	11,000,000,000	10,261	34
2010	S	586,145	58,429,230,250	22,327,500,000	38,092	38
2011	C L	NA	NA	NA	NA	NA

Table 2: Cost of Fertilizer Subsidy under Market Stabilisation Scheme (MSS, 2001-2010) and Growth Enhancement Support Scheme (GESS, 2011-2015)

2012	120,097	13,210,670,000	6,605,335,000	55,000	50
2013	536,095	58,970,450,000	29,485,225,000	55,000	50
2014	1,381,818	152,000,000,000	76,000,000,000	55,000	50
2015	NA	NA	NA	NA	NA

Source-: Ayoola and Ayoola (2016).

IV.2 Challenges of Implementing Agricultural Subsidy Regimes in Nigeria- Empirical Reviews

Arising from the studies by Idachaba, (2014); Oredipe (2015); NAERLS, (2012); Adebo, (2014); Ayoola & Ayoola, (2016); Grow Africa, (2016); Oyediran et al. (2015); FSAN, (2012); Fadairo et al. (2015); Motilewa et al. (2015);); Nwalieji et al. (2013); and Druilhe and Hurlé, (2012), challenges of implementing agricultural input subsidies in Nigeria have been identified as follows:

- 1. Distribution Inefficiencies;
- 2. Lack of Coordination;
- 3. Information Asymmetry/Principal Agent Problem;
- 4. Poor Farmer-Data Management;
- 5. Low Coverage/Enrolment;
- 6. Delay in delivery/distribution of Inputs;
- 7. Inadequate Manpower;
- 8. Illiteracy and Poor ICT Skills among Farmers;
- Engagement of many inexperienced agro-dealers who lack formal training on the products they sell and are unable to provide effective extension services to their customers;
- 10. Elite Capture due to the interference of influential people;
- 11. Supply and/or Delay of inappropriate inputs in which case the inputs delivered may be sub-standard or counterfeit;
- 12. Government Debt/Heavy Fiscal Burden;
- 13. Cost involved, late distributions of vouchers causing farmers not to use basal fertilizers, farmers accepting small amounts of money in exchange of the vouchers instead of buying inputs, thefts of agricultural inputs vouchers, farmers resistance to use of inputs chosen by the government, and financial inabilities of some farmers to copay the price as challenges facing agricultural input subsidy programs in Sub-Saharan Africa;
- 14. Creation of dependency (entitlement) mentality whereby farmers now expect input subsidies as a right;
- 15. Provision of input subsidies leading to agricultural policy neglect in other areas;
- 16. Corruption in the fertilizer subsidy programme caused by government monopoly of fertilizer procurement and distribution; and

168 Central Bank of Nigeria Economic and Financial Review De

December 2021

17. Protection of inefficient farmers from prevailing price realities and the misallocation of resources.

V. Agricultural Subsidies from an International Perspective

V.1 Agricultural Subsidy in the US

Agricultural subsidy regime in the US follows a five-year cycle based on government reviews of the food and farm landscape and renewal of the Farm Bill. The current five-year farm bill is US\$867 billion package titled the Agricultural Improvement Act of 2018. The Bill addresses agricultural and food policy through a variety of programmes relating nutrition assistance, crop insurance, commodity support, and conservation. The 2018 Farm Act is US\$1.80 billion (less than 1.0 per cent) higher than the level projected for a continuation of the previous farm act for FY2019-FY2023 spending. The Congressional Budget Office projects that 76.0 per cent of outlays under the 2018 Farm Act funds nutrition programmes, 9.0 per cent funds crop insurance programs, 7.0 per cent funds conservation programs, 7.0 per cent funds conservation programs, 7.0 per cent funds he remaining 1.0 per cent funds all other programs, including trade, credit, rural development, research and extension, forestry, horticulture, and miscellaneous programs (ERS, 2018).

There is also Environmental Quality Incentives Program (EQIP) subsidy under the 2018 farm Act that provides financial and technical assistance to agricultural producers to address natural resource concerns and deliver environmental benefits such as improved water and air quality, conserved ground and surface water, increased soil health and reduced soil erosion and sedimentation, and improved or created wildlife habitat (Benefits.gov).

V.2 Agricultural Subsidy in China

The OECD's agricultural support estimates indicate China has the highest agricultural subsidy rate, measured as the percentage of GDP, of any of the major agricultural-subsidy countries, except South Korea (OECD 2019). China's agricultural policy has undergone fundamental changes in the last decades. Concerned about national food self-sufficiency and rural household incomes, China changed its longstanding policy of taxing farm households and began instead to subsidise them. This shift has been facilitated by a transformation of the economy and strong economic growth performance in the last three decades, particularly in the non-agricultural sector. This structural transformation in agricultural policies in China conforms with the pattern experienced by other developed countries in the past: as agriculture plays a smaller role in the economy, they shift from a rural to an urban manufacturing and service economy.

China's admission into the World Trade Organisation seems to have had a discernible impact on agricultural subsidy rates that could not be explained by other internal political-economic factors. Accordingly, China's changing agricultural policy can be situated within the national, ideological, and political factors so that the determinants of subsidy rates can be better comprehended. It has been opined that in the next decade, Chinese levels of support to agriculture will mimic those of the United States and other developed countries as measured by producer subsidy equivalents. In spite of the increased levels of support to domestic farmers, even under the WTO, China will remain a potentially attractive market for countries exporting agricultural products, particularly animal products, where subsidies are predicted to remain at lower levels than those targeting crop farmers, and as Chinese diets continue to shift toward animal products. China has been able to attain impressive growth because of the early economic reforms. The reforms have improved economic efficiency in agricultural production, processing, and marketing.

V.3 Agricultural Subsidy in Malawi

In the case of Malawi, the introduction of agricultural input subsidies were able to transform the agricultural sector from a food deficit to that of food security. Some of the highlights of the subsidy regime in Malawi included:

Direct administration of fertiliser subsidies through various regional and district outlets where coupons are distributed to beneficiaries in all administrative areas in the country.

Chirwa et al. (2011) revealed that the main goal of the Farm Input Subsidy Programme (FISP) in Malawi is to raise incomes and household food security of up to two million (out of 3.4 million) smallholder farmers through improvements in their agricultural productivity. The programme targets smallholder farmers who have land but cannot afford to purchase inputs (principally maize seed and fertilisers) at market prices. In line with this, Holden and Lunduka (2013) submit that low use of agricultural inputs in Malawi is primarily caused by limited ability to buy inputs and not time-inconsistent behaviour. They recommended that the current input subsidy design in Malawi should be replaced by smarter and more cost-effective designs that involve smaller packages of fertilizer and delivery of inputs at harvest time, as well as at planting time. Dorward and Chirwa (2011) reveal that the use of voucher as smart subsidy had similar shortcomings just like the universal subsidy programme. Similar findings were also observed by Holden and Lunduka (2013) in Malawi, where a subsidy programme aimed to provide coupons for purchase of subsidised fertiliser and seeds targeted at poor rural households also faced serious problem. The critical findings were that the poverty and vulnerability reduction potentials of the programme were not optimal, leakages of coupons and fertilizers and misallocation of coupons away from the needy resulted through rent-seeking.

Chirwa et al. (2011), working on conceptualising graduation from agricultural inputs subsidies in Malawi, considered ways in which the concept of graduation may be usefully applied to the Farm Input Subsidy Programme (FISP). They set out a broad conceptualisation of graduation for potential application in programme design and implementation. For the Malawian farmers to graduate from relying on fertiliser subsidy and be able to purchase fertiliser at competitive price, Chirwa et al. (2011) recommended potential graduation conditions which include reduced input prices, substitution with cheaper inputs, increased working capital for input purchases, diversification out of maize production, and access to low cost credit for input purchases.

V.4 Lessons Learnt from International Best Practices for Effective Agricultural Subsidy Implementation

- (i) Research and Development (R&D) in the US The Agriculture Improvement Act of 2018 (2018 Farm Act) establishes the Agriculture Advanced Research and Development Authority (AGARDA) pilot authority to develop technologies, research tools, and products through advanced research on long-term and high-risk challenges for food and agriculture. AGARDA focuses on R&D that private industry is unlikely to undertake. This research will help ensure that the United States maintains its position as a leader in global agricultural R&D (Rubenstein, 2019). New high-priority research and extension initiatives focus on: fertilizer management, nutrient management, dryland farming, hop plant health, and coordination of pollinator research. Others include: mechanisation for labour-intensive tasks; removal of barriers to entry for young, beginning, socially disadvantaged, veteran, and immigrant farmers and ranchers; soil health; collaboration with biomedical researchers; size-controlling rootstock systems, invasive species, natural enemy complexes, soil microbiome, pesticide applications, and drift prevention, systems to improve storage life, greater mechanisation, and pest management. Although these focus areas may not directly impact agricultural subsidies, but they will no doubt impact food security.
- (ii) Unmanned Aerial Vehicles (UAVs) for Inspection in South Korea South Korea is one of the highest subsidies providing country, where government support accounts for 54.0 per cent of farmers' income (Park & Park 2015). South Korean local agriculture has been greatly dependent on government subsidy to compete with cheaper imported products and to shift to modern farming practices. However, in response to risks of corruption, fraud, poor internal control procedures and lack of detailed

information, the Korean National Agricultural Products Quality Management Service (NAQS) launched an inspection programme for preventing fraud and securing proper use of government money. The inspection is carried out through site visits and interviews with a sample of the selected application farm. The increasing target area of implementation creates a huge work pressure to the limited numbered investigators. Besides, difficult in accessibility of mountainous areas is also making the job harder. Therefore, it generates more misuse of manpower and government's resources with minimum outcome. Inspection through remotely sensed image can greatly improve the whole system of inspection saving more time and money. Unmanned Aerial Vehicles (UAV) borne image can be used as an alternative actual site visit (Park & Park, 2015). It is advantageous than satellite and aircraft in terms of accessibility, repeatability, and timeliness (Zhang & Kovacs, 2012). In particular, the timeliness is essential to identify crops with a short growing cycle (Park, 2015).

- (iii) Efficient Targeting in India In the agriculture sector, one of the challenges faced is to increase the quick delivery of agricultural subsidies and ensures accurate targeting of the beneficiaries. In this regard, in January 2013 the Government of India launched the direct cash transfer scheme under direct benefit transfer for fund distribution and delivery system. In this scheme, the subsidies (cash/payments) are directly credited to the farmer's account and validated by the Unique Identification Authority of India (UIDAI) to bring transparency, reduce fraud and prevent delays by the hierarchy of government administrative offices (Bakare et al., 2021).
- (i6) Smart Subsidies in East Africa Evidence from some East-African countries suggests that the smart subsidy initiatives have largely succeeded in increasing productivity, production, incomes, and food security (Wiredu, 2015). Smart subsidy is a new form of subsidy programmes which is expected to improve access and use of farm inputs, increase agricultural production and productivity, and ultimately improve the well-being of farmers (Wiredu, 2015). The concept is based on the economic principles of efficiency, equity, and sustainability (Baltzer & Hansen, 2011). Hence, smart subsidies are provided to specific targets (beneficiaries) over a given period with measurable impacts and achievable goals. Smart subsidy instruments include vouchers, targeting, rationing, loan guarantees, demonstration packs and matching grants (Dorward & Chirwa, 2014) built-in to safeguard against fraud (Gregory, 2006).Smart subsidies are expected to adhere to a number of its cardinal design principles which are, its proport targeting, having market-based solutions (based on demand and

172 Central Bank of Nigeria Economic and Financial Review

December 2021

supply), the active participation of the private sector, and implemented within some defined period having a credible exit strategy to put a time limit on the support (Baltzer & Hansen, 2011; Minde et al., 2008; Tiba, 2009; Dorward, 2009; Chirwa & Dorward, 2013).

VI. Paradigm Shift for Agricultural Subsidy for Food Security

VI.1 Refocusing Agricultural Subsidy for Enhanced Food Security in Nigeria

Arising from the discourse in the previous sections, it becomes expedient to provide some guides for refocusing agricultural subsidy for enhanced food security. The discussion is based on the lessons learned from the challenges of agricultural subsidy implementation in Nigeria and the lessons from international experience in agricultural subsidy implementation. It is believed that the suggestions being proposed will help in boosting food security through the different pillars of physical availability, physical and economic access, utilisation and stability. Given that food security has both supply side and demand side implications, it is evident that any efficient and effective agricultural subsidy regime must be holistic and take into consideration the gamut of activities and stakeholders involved in delivering food in the right quality and quantity and at affordable prices over time. Accordingly, the following recommendations for ensuring agricultural subsidies impact positively on food security are being proposed.

Smart Agricultural Subsidy

Smart subsidy is a new form of subsidy programs which is designed to improve access and use of farm inputs, increase agricultural production and productivity, and ultimately improve the well-being of farmers (Wiredu, 2015). According to Baltzer and Hansen (2011), smart subsidies are based on the economic principles of efficiency, equity, and sustainability- hence they are provided to specific beneficiaries over a given period with measurable impacts and achievable goals. Smart subsidy instruments include vouchers, targeting, rationing, loan guarantees, demonstration packs and matching grants (Dorward & Chirwa, 2014) which are built-in to safeguard against fraud (Gregory, 2006).

Focusing on fertiliser subsidies, Morris et al. (2007) suggested ten (10) key features of market-smart subsidies which can be generalised to input subsidies; they are to:

i. Promote input as part of a wider strategy. Interventions designed to promote increased use of inputs should be developed within the context of a wider sector strategy that recognises the importance of supplying complementary inputs, strengthening output markets, and appropriately sequencing interventions.

- ii. Favour market-based solutions. Long-term solutions to the problem of inputs will have to be market-based. Interventions designed to promote increased use of inputs should be designed to support market development and not undermine incentives for private sector investment.
- iii. Promote competition. Competition in input markets is needed to ensure good performance. Barriers to entry into input distribution should be reduced (except possibly in the very short run), and markets should be competitive to ensure the lowest-cost and best quality service.
- iv. Pay attention to demand. Farmers' effective demand, shaped by the current or potential profitability of input use, should be the ultimate driving force of input supply systems and the foundation of a sustainable input promotion strategy.
- v. Insist on economic efficiency. Input promotion efforts should be driven by economic considerations. Interventions designed to promote increased use of input should be carried out only where input use is economically efficient.
- vi. Empower farmers. Farmers should be in the driver's seat. Interventions designed to promote increased use of input should empower farmers to make their own decisions on the most appropriate way to manage soil fertility in their farming context.
- vii. Devise an exit strategy. Governments should not be in the input distribution business for the long haul. Public sector interventions designed to promote increased use of input should be designed with a clear exit strategy, except for a few long-run public-good functions such as market regulation, infrastructural development, and research and development on natural resources management.
- viii. Pursue regional integration. Market size matters, so trade matters. Countries should seek regional integration and harmonisation of input policies to reap economies of size and scope, which are especially important in a region such as Africa with so many small countries.
- ix. Ensure sustainability. Solutions must be designed for the long term. Interventions designed to promote increased use of input should be economically, institutionally, and environmentally sustainable.

Promote pro-poor growth. Equity considerations matter. Assuming that the previous nine guiding principles have been followed, a final consideration is that public interventions designed to promote increased use of input should also aim to promote pro-poor growth. In exceptional circumstances, poverty reduction or food security objectives may even be given precedence over efficiency and sustainability goals, if it can be determined that input interventions are a cost-effective way of addressing these problems.

Distribution of Smart Subsidies- The Role of Identity Management

Nigeria's National identity Management Commission (NIMC) was established to create, operate and manage the nation's identity card database, integrate the existing identity database in government institutions, register individuals and legal residents, assign a unique national identification number and introduce general multi-purpose cards. The digital identification system assigns a unique national identification number and biometric check of uniqueness; issues a national identity (smart) card to each registered individual; provides an identity verification and authentication service infrastructure; and harmonises and integrate existing identification databases in Nigeria (World Bank, 2016a). These functions together with data from the National Social Register (NSR) establishes the uniqueness of everyone using a unique identity number (NIN).

The NSR consolidates information on poor and vulnerable households from states. Also, to ensure that the poor in the State Social Register and NSR are unique, the data therein needs to be linked with biometric profiles of individuals under NIMC. Linking individuals in an SR with the biometric profiles in a national identity registry can improve the accuracy and integrity of information (World Bank, 2016b). This database will help to prevent farmers from registering multiple times to get program benefits and de-duplicate the database to prevent possible leakages and corruption (World Bank, 2016a).

Digital identity can be instrumental in supporting agricultural development. This was employed in the GESS program of the Federal Ministry of Agriculture and Rural Development (FMARD) which provided support directly to farmers to enable them to procure agricultural inputs at affordable prices, and at the right time and place. The identities of the farmers were verified from the NIMC database to eliminate the problem of multiple registrations by beneficiaries (Amurtiya et al., 2018). Under GESS program, fertilisers and seeds are made available to farmers at subsidised prices to encourage agricultural activity.

Effective Targeting

Universal fertilizer subsidies (i.e. subsidies for all) are socially regressive because they create rents for better-off producers who would have used fertilizers anyway; the result is known as "displacement" which refers to the non-subsidised sales that are displaced as a result of the subsidy, and in worse case scenarios generate no increment in total fertilizer use. Experience with universal subsidies in SSA, according to Morris et al. (2007) indicates that it was largely negative, as it resulted in inefficiencies, such as adverse selection of programme beneficiaries i.e. elite capture and displacement of commercial sales, and had disproportionate fiscal costs against their benefits.

Universal fertiliser subsidies are socially regressive because they create rents for better-off producers who would have used fertilisers anyway; the result is known as "displacement" which refers to the non-subsidised sales that are displaced as a result of the subsidy, and in worse case scenarios generate no increment in total fertilizer use (Druilhe and Barreiro-Hurlé, 2012).

In this era of constrained government budgets, development programmes that are targeted towards specific populations with specific policy goals can maximise 'poverty removal benefits accruing from a given burden of cost' (Sen, 1995). To ensure effective targeting in terms of distribution and collection of subsidies, agricultural subsidy programmes should be implemented in accordance with operational guidelines that focus and reach only the intended population as the success of an intervention lies in how it is effectively targeted and achieves stated objectives.

Participatory Subsidy Regimes

Central to the identification of eligible programme participants is the issue of asymmetric information (Ravallion, 2003). While targeted programmes are designed to reach only those who meet certain eligibility criteria, it is rarely, if ever, possible for central administrators to know precisely who meets the criteria at the local level. One of the key arguments in favour of a decentralised targeting approach, such as community-based targeting (CBT), has been based on its potential to identify potential programme beneficiaries accurately by drawing on local knowledge and preferences that might otherwise be unknown to the programme administrators at the central level (Mansuri & Rao, 2012). This assumption propels decentralisation as a potential solution to the asymmetric information problem. However, as Conning and Kevane (2002) aptly explain in the context of decentralised targeting strategies, one of the challenges is to maximise stated welfare objectives subject to the constraining effects of local elite capture.

Information symmetry is key to participatory development, which involves including people who are affected by development processes as planners and implementers (Thomas, 2013). Participatory development broadens the participation linkages to mean not only the top governments and donor agencies) on one hand and the bottom (rural communities who are the beneficiaries) but also those who may not belong to the two groups but were affected by the developmental processes.

Holistic Approach to Agricultural Subsidy

Agricultural Sector is divided into four fundamental sub-sectors, specifically, Crops, Livestock, Aquaculture and Forestry. However, most ground-breaking agricultural subsidy schemes implemented in Nigeria to ensure food security particularly Economic and Financial Review

December 2021

focused the crop sub sector. In 2020, the agricultural sector contributes about 24.0 per cent to the national GDP. The crop sub sector contributed about highest (90.0 per cent) of the sector GDP, fisheries 4.0 per cent, livestock about 5.0 per cent while the forestry contributes the least (1.0 per cent) (statista.com). However, potentials exist for other agriculture subsectors to contribute more to gross national product if right incentive are provided and properly harnessed holistically. Notwithstanding, the forest subsector plays a major role in providing industrial raw materials for industries, income in form of employment generation and energy supply for domestic and industrial use (Odetola & Etumnu, 2013). In addition to the above, fishery and livestock sub sectors provide the country with the cheapest animal protein source especially among the rural dwellers in Nigeria. In Nigeria, forest is managed by the government which makes no provision for incentives for stakeholders to invest in sustainable forest management which has resulted in uncontrolled deforestation of the natural forest and desert encroachment (Olaoye & Ojebiyi, 2018). A holistic agricultural subsidy which incorporates condition farm financial aid on the protection of forests will not only conserve the natural biodiversity but will also encourage effort to produce more thereby increases agricultural profitability and provide incentive to increases food production on existing land thus food availability.

Climate Smart Subsidy

Climate-smart subsidy is an integrated approach to managing landscapes cropland, livestock, forests, and fisheries that addresses the interlinked challenges of food security. A growing global population and changing diets are driving up the demand for food. Production is struggling to keep up as crop yields level off in many parts of the world, ocean health declines, and natural resources including soils, water, and biodiversity are stretched dangerously thin. A 2020 report found that nearly 690 million people or 8.9 per cent of the global population are hungry, up by nearly 60 million in five years. The food security challenge will only become more difficult, as the world will need to produce about 70 percent more food by 2050 to feed an estimated 9 billion people (World Bank, 2020). Therefore, to both feed the world and solve climate change, the world needs to produce 50.0 per cent more food in 2050 compared to 2010 while reducing greenhouse gas emissions by twothirds. However, studies show that agricultural subsidies are currently doing little to achieve these goals as only 5.0 per cent of this funding supports any kind of conservation objective (Searchinger, 2020; Salman 2019). Going forward, climate smart policies to maintain and enhance soil fertility are as follows:

- i. review GES subsidy and align with overall fertiliser supply;
- ii. crop rotation to improve nitrogen fixation;
- iii. soil mapping and testing;
- iv. soil fertility reconstruction and formal fallow periods;

177

- v. fertiliser quality control;
- vi. use of organic fertiliser;
- vii. crop specific fertiliser formulation;
- viii. erosion control measures; and
- ix. improved conservation, reforestation, and green belt policies.

Agricultural Subsidy should Recognise Research-Extension-Farmers-linkage System for Food Security

One important way of supporting food security is through agricultural extension programmes. Agricultural extension is broadly defined as the process of development of agricultural knowledge and skills amongst farmers, aimed at increasing their productivity and realising other desirable changes (Collett & Gale, 2009). Agricultural extension has multiple goals, including transferring knowledge from global, national, and local researchers to farmers, helping them clarify their own goals and assessing their opportunities, educating them about decision-making processes, and promoting desirable agricultural development (Msuya et al., 2017).

Agricultural extension plays a role in improving farmers' productivity and incomes, thereby reducing poverty, and increasing food security (Raidimi & Kabiti, 2019). If farmers are producing cash crops, increased productivity provides a pathway of earning income which can be utilised for food procurement (Conceição et al., 2016). Furthermore, increased productivity of food crops ensures increased availability of household food. The productivity of farmers is increased through application of appropriate knowledge in response to production challenges such as pests, diseases, and changes in weather patterns (FAO, 2017). Agricultural extension can play a role by ensuring that efforts towards increasing productivity are sustainable. Sustainable agricultural production ensures that current production activities do not compromise the production chances in the future.

Empowering farmers as key agents of change while fostering co-creation of knowledge, integrating traditional, practical, and local practices and skills is required for sustainable production (FAO, 2018). An increase in farmers' incomes is achieved through market information and linking farmers to markets. Introduction of agricultural expertise, farmer responsive plans, feasible credit scheme, free land distribution and effective extension programs can improve the yield of cereals, fruits, vegetables, and animal related products (Fiaz et al., 2018). Self-sufficiency in food production could only be gained by addressing the agricultural problems and keeping the farmers aware of modern agricultural technologies, necessary for improving productivity, by the effective use of extension services. Agricultural extension is considered as a service to spread/extend information based on

178 Central Bank of Nigeria Economic and Financial Review December 2021

research, to the rural people to develop their living standard (IFPRI, 2010). Hence, providing incentives in form of subsidies to address Research-Extension-Farmer-Input-Linkage-System as done in the past can help boost food security.

By developing their farming system and having access to successful agricultural extension services, the farmers have been able to improve their production efficiency for all major agricultural products to assure optimum food production. For example, the important agricultural extension measure of rice production in Vietnam was "Three Reductions, Three Gains" project that aimed to reduce the production cost by reducing the amount of seed, fertilizer, and pesticides, whilst gaining a higher yield, better rice quality, and more profit (Sattaka et al., 2017).

Value Chain Responsive Subsidy and Food Security

Input subsidies is essential at removing the difficulties in production by giving access to important inputs, such as seeds and fertilizers to farmers who would otherwise not be able to afford them. It is often argued that input subsidies can be a temporary "bridge" towards self-sustaining input application by increasing household income, allowing households to purchase inputs in the future (Ellis & Maliro, 2013; Bizikova et al., 2020). When inputs are subsidised, it has multipliers effect as there will be increase in production of farmers thereby increasing their income. When the income of the farmers is increase, it provides more opportunity for the famers to spend more on their essential needs, they can but more for foods they don't produce.

According to Ellis and Maliro (2013) and Kaplan et al. (2016) food vouchers is also an important element that can supports households to overcome barriers to the uptake of subsidised fertilizer, thereby contributing to food security. This suggests that food vouchers/cash transfers are very effective as complementary strategies when combined with input subsidies, particularly to bridge periods of food scarcity. Rutherford et al. (2016) and Cleaver (2013) find that value chain interventions generally increase the quality and availability of foods. The promotion of staple food value chains contributes to increased production and falling prices, which are particularly effective in building food security (Kaplan et al., 2016).

Monitoring and Evaluation with Respect to Tracking Subsidy Distribution, Collection, and Use

Effectively delivering agricultural subsidies will require monitoring and evaluation (M&E). This will include baseline assessments, indicators for assessing the impacts of interventions, and capacity requirements. While monitoring measures aspects of project implementation and aims to improve the project's design and function along the way, evaluation studies the outcomes of the project (e.g., improvements

in food production, uptake of new agro-technologies, changes in farmers' income, cost-effectiveness, etc.) with the aim of informing the design of future programmes.

Monitoring and evaluation are about identifying and measuring the drivers of success just as much as understanding the causes of failure. Effective M&E incurs costs, and these need to be identified upfront, budgeted for, and ring-fenced for this purpose, so that they are not lost through oversight (or over-spending) as the programme/policy is implemented. The uncertainty of the likely effects of climate change on agriculture adds a further challenge.

It is also vital that the subject of M&E has a schedule of pre-agreed deliverables and outcomes, which must be specific and measurable. In this case policies, programmes, projects and activities should be set out in the agricultural subsidy framework. Assessment of the outcomes within this framework will highlight economic, environmental, and social impacts. The schedule for monitoring and evaluation also needs to be identified and planned right at the beginning of the project cycle, and M&E needs to be integrated into the full project cycle through regular planning processes. Failure to undertake this planning at the initial stage of the project cycle will make M&E ineffectual down the line as the project develops.

Furthermore, effective M&E needs the involvement of key stakeholders, and these need to be identified early on. A combination of a top-down analytical approach and a bottom-up approach featuring community involvement/engagement is strongly recommended by all the key institutions. The challenge here is how to integrate participatory and non-participatory monitoring and evaluation, given the diverse range of stakeholders. The use of Unmanned Aerial Vehicles for subsidy monitoring as done in South Korea can help substantially in reducing fraud and corruption while ensuring real time and up-to-date information on subsidy.

Public-Private Partnership

When markets for the delivery of innovation fail, Governments may consider different policy alternatives, such as public investment, policy instruments such as subsidies or tax incentives, and a PPP approach. The decision for the government to adopt a PPP approach to pursue a given objective should be guided by the balance of costs and benefits, compared with other alternatives, as participation in PPPs should be for all innovation actors. The fundamental rationale for public and private actors to join forces in a PPP arises when individual actors alone cannot produce the same service or output or do it at a higher cost. For both the public and private sectors, the benefits from PPPs come from the pooling of resources and the complementarily of capacities, while risks and costs are reduced because they are shared. Compared to subsidies, PPPs help avoid impeding further development, and allow for the development of integrated solutions.

180 Central Bank of Nigeria Economic and Financial Review December 2021

The rationale for PPPs in innovation also includes coordination failures, although other policy instruments could be used to address them (e.g. tax credits for collaboration). The general rationale for public intervention in innovation such as skills development, network development, or the ability to join in global innovation networks are also cited as reasons for PPPs in this area (OECD, 2014). PPPs help build innovation capabilities, improve connectivity between national innovation systems and provide compatible incentives to all stakeholders. Fostering links and understandings between public and private researchers, but also between the research and business/farm community, can be particularly useful in:

- lowering transaction costs between players, facilitating thus future cooperation.

- improving knowledge transfer, where a smaller organisation or country lacks capacity to fund the critical mass necessary to enable spill-in of knowledge.

Furthermore, during the GESS, a private sector channel for input distribution was developed with the primary objective of depoliticising the input sector, this was done by disallowing States from procurement of inputs (Grow Africa, 2016). A scaling-up on this could be explored.

Subsidy Distribution-Right Timing and Quantity

Given that most agricultural activities are seasonal in nature, there is a need for timely delivery and disbursement of inputs. Studies have shown that late delivery of inputs is a major challenge to the successful implementation of subsidy programmes across the country (Federal Ministry of Agriculture and Rural Development, 2016; Nwaobiala, 2015; Oyediran et al., 2015; Fertiliser Suppliers Association of Nigeria, 2012).

Also, Government's inability to release funds to agro-dealers in good time has resulted in late or even non-delivery of inputs in some cases. This has implications for the input suppliers who supply products on credit (Fertiliser Suppliers Association of Nigeria, 2012).

The trend of leadership in the country had led to inconsistencies and instability in fertiliser subsidy polices; and the gains are not widely spread among the targeted beneficiaries –thus having a negative implication on the increased food production programme.

Mode of Redemption of Agricultural Subsidy

Despite innovations in the design and implementation of fertiliser subsidies, one significant pitfall is "vote-buying" where farmers collect vouchers, they have no intention of using or could not afford to use (Banful & Olayide, 2010). They rightly

predicted that there would be periods of shortage of vouchers and sold the vouchers to other farmers who desperately needed to apply fertiliser.

Banful and Olayide (2010) opined that the parallel sales of subsidised and market fertiliser (unsubsidised) in Nigeria tend to create an avenue for lower-priced subsidised fertilisers to be diverted for sale at higher market prices. These shortcomings of fertiliser subsidies led to introduction of vouchers or smart subsidies or coupons. Thus, to achieve food security, there is need to identify forms by which subsidies will be redeemed.

VII. Conclusion

This paper focused on refocusing agricultural subsidy for food security in Nigeria. The paper identified the pathway of agricultural subsidy for impacting on food security. A review of agricultural subsidy in Nigeria reveals the key constraint to include poor database, lack of coordination, corruption, poor targeting, and poor distribution system. The review of international experience shows that research on new technology, effective targeting, smart agricultural subsidy, monitoring through unmanned aerial vehicles are *sine qua non* for effective agricultural subsidy implementation with a possible positive impact on food security.

Based on the lessons from both Nigeria and international experiences on agricultural subsidy; the paper recommends the following possible options for effective implementation: smart agricultural subsidy, proper identity management, participatory subsidy regime, extension of subsidy to other sectors beyond crops, climate-smart agricultural budget and broad subsidy across the value chain. Other effective agricultural subsidy, appropriate subsidy redemption mechanism and due timing in implementing agricultural subsidy possibilities will help improve food security by ensuring availability of food, enhanced physical and economic access to food, improved food utilisation and sustained stability of food security in Nigeria.

References

- Adebo, G. M. (2014). Effectiveness of e-wallet practice in grassroots agricultural services delivery in Nigeria: A case study of Kwara State growth enhancement support scheme. Journal of Experimental Biology and Agricultural Sciences, 2(2320), 410–418.
- Afolayan S. O. (1997). Effect of irrigation frequency on soil moisture potential and fruit yield of Okra (Abelmoschuss esculentus) paper presented at the 15th annual conference of HORTSON held at NIHORT, Ibadan April 8-11, 1997.
- Amurtiya M., & Abdu K. T., & David C. M. (2018). Agricultural inputs subsidy in Nigeria: An overview of the growth enhancement support scheme (GESS).
 Acta Universitatis Agriculturae et Silviculturae Mendelianae Brunensis, Mendel University Press, 66(3), 781-789. Assistance and Nutrition Research Report 11-2.
- Ayoola, G. B. (2001). Essays on the agricultural economy 1: A book of readings on agricultural development policy and administration in Nigeria. Ibadan: TMA Publishers.
- Ayoola, G., & Ayoola, J. S. (2016). The transformation of smallholder agriculture in Africa: Lessons in policy experimentation and governance from Nigeria. Addis Ababa, Ethiopia.
- Bakare S., Shindez S., Hubballi R., Hebbale G., & Joshis V. (2021) A blockchain-based framework for Agriculture subsidy disbursement. IOP Conf. Series: doi:10.1088/1757-899X/1110/1/012008.
- Baltzer, K. & Hansen, H. (2011) Agricultural Input Subsidies in Sub Saharan Africa.Evaluation Study 2011/2, Ministry of Foreign Affairs Denmark. Retreived from: http://www.oecd.org/derec/49231998.pdf.
- Banful, A. B., & Olayide, O. (2010). Perspectives of varied stakeholders in Nigeria on the federal and state fertilizer subsidy programmes. Nigeria Strategy Support Programme Working Paper Number 08. International Food Policy Research Institute, Washington, D. C.
- Benefits.gov. What is environmental quality incentives program? https://www.benefits.gov/benefit/375.
- Bizikova, L., Jungcurt, S., McDougal, K., & Tyler, S. (2020). How can agricultural interventions enhance contribution to food security and SDG 2.1? Global Food Security, 26 (2020) 100450.
- Chirwa, E., & Dorward, A. (2013). The role of the private sector in the Farm Input Subsidy Programme in Malawi. Available at: http://www.futureagricultures.org/publications/research-and-analysis/1743-the-role-of-theprivate-sector-in-the-farm-input-subsidy-programme-inmalawi/file.
- Chirwa, E. W., Matita, M. M., Mvula, P. M., & Dorward, A. R. (2011). Impacts of the farm input subsidy programme in Malawi. Paper prepared for Malawi

Government/ DFID Evaluation of Malawi Farm Input Subsidy Programme. School of Oriental and African Studies, University of London.

- Cleaver, K., (2013). The importance of scaling up for agricultural and rural development: A success story from Peru. IFAD Occasional Paper. https://www.ifad.org/documents/10180/ae15c993-f5b7-4ab1-aaf8-800f1b59420a.
- Collett, K., & Gale, C. (2009). Training for rural development: Agricultural and enterprise skills for women smallholders. City and Guilds Centre for Skills Development Report. London, England.
- Conceicao, P., Levine, S., Lipton, M., & Warren-Rodríguez, A. (2016). Toward a food secure future: Ensuring food security for sustainable human development in sub-Saharan Africa. *Food Policy*, 60,1-9.
- Conning, J., & Kevane, M. (2002). Community-based targeting mechanisms for social safety nets: A critical review. World development, 30(3), 375-394.
- Dorward, A. (2009). Rethinking agricultural input subsidy programmes in a changing world. London, UK: School of Oriental and African Studies, University of London.
- Dorward, A., & Chirwa, E. (2014). The rehabilitation of agricultural input subsidies? London, UK. London, UK: School of Oriental and African Studies, University of London.
- Dorward, A., & Chirwa, E. (2011). The Malawi agricultural input subsidy programme: 2005/06 to 2008/09. International Journal of Agricultural Sustainability, 9(1), 232–47.
- Druilhe, Z., & Barreiro-hurle, J. (2012). Fertilizer subsidies in sub-Saharan Africa. FAO ESA working paper No 12 – 04. Rome, Italy: FAO.
- Eboh, E. C., Ujah O. C., & Amaechina E. C. (2006). Do government fertilizer subsidies benefit rural poor farmers in Nigeria? Making sense and of existing data. A Paper Presented during the 5th PEP Research Network General Meeting June 18-22, 2006. Addis – Ababa, Ethiopia.
- Economic Research Service (ERS), (2018). Agriculture Improvement Act of 2018: Highlights and Implications. U.S. Department of Agriculture. Retrieved from https://www.ers.usda.gov/agriculture-improvement-act-of-2018-highlightsand-implications/.
- Ecker, O., & Breisinger, C. (2012): The food security system; A new conceptual framework. IFPRI Discussion Paper 01166. International Food Policy Research Institute, CGIAR.
- Ellis, F., & Maliro, D. (2013). Fertiliser subsidies and social cash transfers as complementary or competing instruments for reducing vulnerability to hunger: The case of Malawi. *Dev. Pol. Rev.*, *31*(5), 575–596.
- Fadairo, O. S., Olutegbe, N. S., & Tijani, A. M. (2015). Attitude of crop farmers towards e-wallet platform of the Growth Enhancement Support scheme for input

December 2021

delivery in Oke-Ogun area of Oyo state. Journal of Agricultural Informatics, 6(2), 62–71.

- Fan, S., Johnson, M., Saurkar, A., & T. Makombe. (2008). Investing in African agriculture to halve poverty by 2015. IFPRI Discussion Paper 00751. Washington, D.C.: IFPRI.
- Federal Ministry of Agriculture and Rural Development (FMARD), (2016). The agriculture promotion policy (2016–2020) building on the successes of the ATA. Closing Key Gaps Policy and Strategy Document. Federal Ministry of Agriculture and Rural Development Report. Available at: http://fscluster.org/sites/default/files/documents/2016-nigeria-agric-sector-policy-roadmap_june-15-2016_final1.pdf.
- Federal Ministry of Agriculture and Rural Development (FMARD), (2011). Federal Ministry of Agriculture and Rural Development of Nigeria. Agricultural Transformation Agenda: We Will Grow Nigeria's Agricultural Sector.
- Fertilizer Suppliers Association of Nigeria, (FSAN), (2012). The growth enhancement support scheme (GESS) monitoring report 2012. Abuja, Nigeria: FSAN.
- Fiaz, S., Noor, M. A., & Aldosri, F. O. (2018). Achieving food security in the Kingdom of Saudi Arabia through innovation: Potential role of agricultural extension. *Journal of the Saudi Society of Agricultural Sciences*, 17, 365–375.
- Food and Agriculture Organisation (FAO), (2008). An introduction to the basic concepts of food security EC-FAO Food Security Programme, FAO Rome. Retreived from: http://www.fao.org/3/al936e/al936e.pdf.
- Food and Agriculture Organisation (FAO), (2017). The future of food and agriculture – Trends and challenges. FAO Rome. Retrieved from https://www.fao.org/3/i6583e/i6583e.pdf.
- Food and Agriculture Organisation (FAO), (2018). Transforming food and agriculture to achieve the SDGs: 20 interconnected actions to guide decision-makers. FAO Rome. Retrieved from https://www.fao.org/3/19900EN/i9900en.pdf.
- Gautam, M. (2015). Agricultural subsidies: Resurging interest in a perennial debate. Indian Journal of Agricultural Economics, 70(1), 83-105.
- Gautam, M., & Kar, A. (2014). Public expenditure priorities for competitiveness: Farm level evidence on the effectiveness of subsidies in Sri Lanka. The World Bank, Washington, DC.
- Gregory, I. (2006). The role of input vouchers in pro-poor growth. Selected background paper Prepared for the African Fertilizer Summit June 9 13, 2006 Abuja, Nigeria.
- Grow Africa, (2016). Fertilizer subsidy reform revives Nigeria's agriculture: Case studies on PPP agricultural investment. Abuja, Nigeria: Grow Africa.
- Holden, S. T., & Lunduka, R. (2013). Who benefit from Malawi's targeted farm input subsidy program? Forum for Development Studies, 40(1), 1-25.

- Holden, S. T., 2018. "The economics of fertilizer subsidies," CLTS Working Papers 9/18, Norwegian University of Life Sciences, Centre for Land Tenure Studies, revised 16 Oct 2019.
- Ibietan, J. (2011). Public policy making process: Agricultural and rural development policies in Nigeria (1960 to Date). International Journal of Studies in the Humanities, 8(9), 96-108.
- Idachaba, F. S. (1988). Strategies for achieving food self-sufficiency in Nigeria. Key Note Address, 1st National congress of science and technology, University of Ibadan: 16th August, 1988.
- Idachaba, F. S. (1994). Human capital and African agricultural development, 1994 Conference, August 22-29, 1994, Harare, Zimbabwe 183411. International Association of Agricultural Economists.
- Idachaba, F. S. (2014). Government, corruption and the underdevelopment of Nigerian agriculture. 4th Professor Samson Olajuwon Olayide Memorial Lecture, delivered at the University of Ibadan on March, 20, 2014.
- International Food Policy Research Institute (IFPRI), (2010). Agricultural Extension in Ethiopia through a Gender and Governance Lens. Development Strategy and Governance Division. International Food Policy Research Institute – Ethiopia Strategy Support Program 2, Ethiopia.
- Iwuchukwu, J. C., & Igbokwe, E. M. (2012). Lessons from agricultural policies and programmes in Nigeria. *Journal of Law, Policy and Globalisation*, 5, 11-21.
- Jayne, T. S., Mason, N. M., Burke, W. J., & Ariga, J. (2018). Taking stock of Africa's second-generation agricultural input subsidy programs. *Food Policy*, 75, 1-14.
- Kaplan, M., Bettighofer, S., Bruntrup-Seidemann, S., & Noltze, M. (2016). Agricultural value chains. German Institute for Development Evaluation, Bonn, Germany. Available https://www.deval.org/files/content/Dateien/Evaluierung/Berichte/2016_D Eval WSK en final.pdf.
- Li, Y., Zhang, W, Ma, L., Huang, G., Oenema, O., Zhang, F., & Dou, Z. (2013). An analysis of China's fertilizer policies: Impacts on the industry, food security and the environment. *Journal of Environmental Quality*, 42(4), 972-981.
- Liverpool-Tasie, S. O., & Takeshimab, H. (2013). Input promotion within a complex subsector: fertilizer in Nigeria. *Agricultural Economics*, 44, 581–594.
- Mansuri G., Rao V. (2012) Localising development. World Bank Policy Research Report. Retrieved from:

http://siteresources.worldbank.org/INTRES/Resources/469232-

1321568702932/8273725-

1352313091329/PRR_Localizing_Development_full.pdf.

Michael, A., Tashikalma, A. K., & Maurice, D. C. (2018). Agricultural inputs subsidy in Nigeria: An overview of the growth enhancement support scheme (GESS).

186 Central Bank of Nigeria

Acta Universitatis Agriculturae et Silviculturae Mendelianae Brunensis, 66(3), 781-789.

- Minde, I., Jayne, T. S., Ariga, J., & Crawford, E. (2008). Fertilizer subsidies and sustainable agricultural growth in Africa: Current issues and empirical evidence from Malawi, Zambia and Kenya. Food Security Group, Michigan State University. Paper prepared for the Regional Strategic Agricultural Knowledge Support System (Re-SAKSS) for Southern Africa.
- Morris, A. S., Silk, J. S., Steinberg, L., Myers, S. S., & Robinson, L. R. (2007). The role of the family context in the development of emotion regulation. *Social Development*, 16(2), 361–388. doi:10.1111/j.1467-9507.2007.
- Motilewa, B. D., Worlu, R. E. K., Adepoju, Y., & Fayomi, O. O. (2015). Interrogating the citizen-centered services and development in Nigeria's agricultural sector through e-governance. Presented at Covenant University Conference on Egovernance in Nigeria.
- Msuya, C. P., Annor-Frempong, F. K., Magheni, M. N., Agunga, R., Igodan, C., Ladele, A., Huhela, K., Tselaesele, N. M., Msatilomo, H., Chomo, C., & Zwane, E. (2017). The role of agricultural extension in Africa's development: The importance of extension workers and the need for change. International Journal of Agricultural Extension, 5(1), 51-58.
- National Agricultural Extension and Research Liaison Services, (NAERLS) (2012). National Report of Agricultural Performance Survey of 2012 Wet Season in Nigeria. Zaria, Nigeria National Association of Nigerian Traders (NANTS), (2015). E-Wallet research documentary. Agricultural transformation agenda of Mr. President. Available at: www.nants.org/ewallet-research.
- National Bureau of Statistics (NBS), (2020): 2019 poverty and inequality in Nigeria: Executive Summary.
- Ndagwakwa, S. K. (1989). The role of Nigerian agricultural and cooperative bank (NACB) in financing farm production in Yola local government of Gongola State. Unpublished HND project, Ahamadu Bello University, Zaria. P. 69.
- Nwalieji, H. U., Uzuegbunam, C. O., & Okeke, M. N. (2013). Assessment of growth enhancement support scheme among rice farmers in Anambra State, Nigeria. Journal of Agricultural Extension, 19(2): 71–81.
- Nwaobiala, C. U. (2015). Strengthening Agricultural Extension Delivery through Institutional Approach in Nigeria. In: Nwachukwu, I., Onumadu, F. N., Ifenkwe, G. E., Agbarevo, M. B. N., Akpu, U., Odoemelam, L. E., & Nwaobiala, C. U. (Eds.). Contemporary Issues in Extension Systems and Development, 38– 45.
- Odetola T., & Etumnu C. (2013). Contribution of agriculture to economic growth in Nigeria. The 18th Annual Conference of the African Econometric Society (AES) Accra, Ghana at the session organised by the Association for the Advancement of African Women Economists (AAAWE), 22nd and 23rd July, 2013.

- OECD, P. (2014, November). Innovating the public sector: From ideas to impact. OECD Conference Centre, Paris.
- OECD (2019). Producer and consumer support estimates database. Available from URL: http://www.oecd.org/tad/agriculturalpolicies/producerandconsumersupportestimatesda tabase.htm.
- OECD (2021), Agricultural support (indicator). doi: 10.1787/6ea85c58-en (Accessed on 01 September 2021)
- Olaoye, O. J., & Ojebiyi, W. G. (2018). Marine fisheries in Nigeria: A review. Marine Ecology-Biotic and Abiotic Interactions.
- Olisah, C (2021). 2.2 million farmers to benefit from FG's N12.3 billion agricultural subsidy. Retrieved from https://nairametrics.com/2021/07/19/2-2-million-farmers-to-benefit-from-fgs-n12-3-billion-agric-subsidy/.
- Oredipe, A., (2015). Sustaining the Positive Outcome of the Agricultural for Attaining Competitiveness, Sustainability and Inclusive Growth. In: 21st Nigerian Economic Summit. Lagos, Nigeria, pp. 15–30.
- Oyediran, W. O., Dick, T. T., Owolade, E. O., & Oluade, E. A. (2015). Contributions of Growth Enhancement Support Scheme (GESS) programme to food security and poverty alleviation of Agricultural Cooperatives in Ogun State, Nigeria. Journal of Educational Policy and Entrepreneurial Research (JEPER), 2(June): 13–22.
- Ozoani, S. E. (2019). Appraisal of agricultural programmes in Nigeria. International Journal of Innovative Agriculture & Biology Research, 7(1), 86-94.
- Park, J. H. (2015). Study on construction plan of efficient direct payment inspection using unmanned aerial system, National Agricultural Products Quality Management Service, Cheongju, 5-6.
- Park, J. K., & Park, J. H. (2015). Crops classification using imagery of unmanned aerial vehicle (UAV). Journal of the Korean Society of Agricultural Engineers, 57(6), 91-97.
- Raidimi, E. N., & Kabiti, H. M. (2019). A review of the role of agricultural extension and training in achieving sustainable food security: A case of South Africa. S. Afr. J. Agric. Ext., 47(3), 120 130. Retreived from: http://dx.doi.org/10.17159/2413-3221/2019/v47n3a520.
- Ravallion M. (2003). Targeted Transfers in Poor Countries: Revisiting the Trade-Offs and Policy Options.' World Bank Policy Research Working Paper No. 3048. Retrieved from:

http://elibrary.worldbank.org/content/workingpaper/10.1596/1813-9450-3048.

- Ravallion, M. (2009). Evaluation in the practice of development. The World Bank Research Observer, 24(1), 29-53.
- Rubenstein, K. D. (2019). Research, extension, and related matters: Highlights and implications of title vii of agriculture improvement act of 2018. USDA Economic Research Service.

- Rutherford, D. D., Burke, H. M., Cheung, K. K., & Field, S. H., (2016). Impact of an agricultural value chain project on smallholder farmers, households, and children in Liberia. *World Dev*, 83, 70–83.
- Salimonu, K. K. (2007). Attitude to risk in resource allocation among food crop farmers in Osun State, Nigeria. PhD thesis, Agricultural Economics Department, University of Ibadan, Ibadan.
- Salman K. (2019). Political economy of fertilizer subsidy implementation process in Nigeria, Policy Brief; African Economic Research Consortium, No. 633.
- Sattaka, P., Pattaratuma, S., & Attawipakpaisan, G. (2017). Agricultural extension services to foster production sustainability for food and cultural security of glutinous rice farmers in Vietnam, Kasetsart. *Journal of Social Sciences*, 38 (2017), 74-80.
- Searchinger, (2020). Redirecting agricultural subsidies for a sustainable food future. World Resources Institute. Retreived from: https://www.wri.org/insights/redirecting-agricultural-subsidies-sustainable-

food-future.

- Shaw, D. J. (2007). World Food Summit, 1996. World Food Security, 347–360. doi:10.1057/9780230589780_35.
- Statista (2021). Agricultural Subsidies. Retrieved from https://www.statista.com/chart/24416/agricultural-subsidies/.
- Takeshima, H., & Liverpool-Tasie, L. S. (2013). Fertilizer subsidy, political influence and local food prices in sub-Saharan Africa: Evidence from Nigeria. In: Agricultural and Applied Economics Association'sm2013 AAEA and CAES Joint Annual Meeting. Washington DC, USA, 1–38.
- Thomas P. (2013). Challenges for participatory development in contemporary development practice. Development Bulletin No. 75, August. Development Studies Network. Retreived from:

http:crawford.anu.edu.au/rmap/devnet/dev-bulletin.php.

- Tiba, Z. (2009). The role of input subsidies: Operational guidelines on implementation Policies for Good Economic Management of Food Price Swings in African Countries. Rome, Italy.
- Wang, D., Xu, Z., Zhao, J., Wang, Y., & Yu, Z. (2011). Excessive nitrogen application decreases grain yield and increases nitrogen loss in a wheat-soil system. *Acta Agriculturae Scandinavica, Sction B – Soil and Plant Science, 61*(8), 681-692.
- Wilmot, P. F. (1979). Multinational corporation in the third world. A threat to national Independence. Text of a Symposium at the institute of Administration, Ahmadu Bello University, Zaria.
- Wiredu, A. N. (2015). Impacts of fertilizer subsidy on farm-level productivity and food security: A case study of rice-producing households in northern Ghana. An unpublised PhD thesis submited to the Institute of Agricultural Economics and Social Sciences in the Tropics and Subtropics University of Hohenheim.

- World Bank (2014). Tanzania public expenditure review: National agricultural input voucher scheme.
- World Bank (2016a). ID4D Country Diagnostic: Nigeria
- World Bank (2016b). National social safety nets project appraisal document. Report No: PAD1687.
- World Bank. (2020). World Bank Annual Report 2020. Washington, DC: World Bank. doi: 10.1596/978-1-4648-1619-2. License: Creative Commons Attribution– Non-Commercial–No Derivatives 3.0 IGO (CC BY-NC-ND 3.0 IGO). Retreived from:
- www.statista.com/statistics/1207940/share-of-gdp-by-agricultural-sector-innigeria/.
- Zhang, C. and Kovacs, J. (2012). The application of small unmanned aerial systems for precision agriculture: A review. *Precision Agriculture*, 13, 693-7.

Appendices

S/N	Programmes/Subsidy Regimes	Programme Objectives	Implementation Challenges
1	The National Accelerated Food Production programme 1972 (Ozoani, 2019).	 a. To increase food production in the country through education of farmers on productive farming practices. b. To provide credits and inputs to farmers cooperatives to enhance food productivity. 	 a. Farmers financially sponsored the last two phases of the programme which discouraged some farmers from participation. b. Farmers who could not form co-operatives were left out in the programme since the programme relied on disbursement of credits and farm inputs through co-operative societies. c. Premature withdrawal of funding by the Federal Government due to the introduction of Operation Feed the Nation.
2	Agricultural Development Programme 1975 (Ayoola, 2001)	 a. To provide agricultural inputs subsidy to peasant farmers b. To increase agricultural productivity by sustaining domestic food supply c. To provide extension services, technical input support and rural infrastructure to rural farmers 	 a. Shortage of fund due to decline in oil prices which delayed projects implementation. b. Emphasiss is more on sole cropping while majority of the farmers practiced mixed cropping c. Untimeliness of subsidised input supply for the programme
3	Operation Feed the Nation, 1976 (Wilmot, 1979)	 a. To make every available piece of land in urban, sub-urban and rural areas planted. b. To actively involve every discipline in farming activities c. The federal government to provide farmina 	 a. Every available piece of land was farm irrespective of its suitability for agriculture. b. Majority of the participants in the programme had little or no farming background and there was no

Table 2: Programmes with Subsidy Components in Nigeria

	Di cu Duciu	inputs at subsidised rate to individual. d. To bring about increased food in the entire nation	preparatory teaching on farming practices. c. Nepotism in inputs supply d. There was less demand for the food because many people produced partly or wholly food, they consumed
4	Aver Basin Development Authorities (1976) (Ayoola, 2001).	 a. To provide irrigation facilities through dam construction for all year farming. b. To provide hydroelectric power and domestic water in the rural areas for increased agricultural production. c. Provision of rural infrastructural development to stem the rural-urban migration and boost agricultural production. 	 a. Substantial public funds were wasted to streamline sizes and functions of programme through the disposal of non-water assets b. Operations of the authority suffered intensive political interference. c. lack of continuity in commitment to policy
5	Green Revolution 1980 (Iwuchukwu & Igbokwe, 2012)	 a. To provide agricultural inputs and credit facilities to farmers to increase agricultural productivity. b. To subsidize farm mechanisation to peasant farmers. c. To formulate favourable pricing policy for the agricultural products 	 a. The programme suffered delay in execution of most of its projects. b. Lack of proper monitoring and evaluation of the projects. c. Choice of inappropriate organisational structure for implementation of policies;
6	Nigerian Agricultural and Co-operative Bank 1983 (Ndagwakwa, 1989).	 a. To deliver credit for the development of agriculture and other agro-allied industries, including marketing of agricultural products b. To assist and facilitate adoption of modern agricultural technologies and good management practices 	 a. Strict loan policy which bar smallholder farmers' loan accessibility. b. Lack of adequate data for decision making.

December 2021

7	Directorate for Food Roads and Rural Infrastructure, 1986. (Idachaba, 1988).	 a. To improve the quality of life and living standard of the rural dwellers b. To actively engage the rural people in rural infrastructural development 	 a. The programme suffered fund mismanagement and highly criticised for lack of proper focus and accountability b. The programme recorded poor quality of infrastructure due to fund embezzlement and mismanagement.
8	National Fadama Development Project, 1990 (Afolayan, 1997)	 a. To promote simple low- cost improved irrigation technology b. To sustainably increase the incomes of the fadama users through expansion of farm and non-farm activities with high value-added output 	a. Unskilled handling of water application through irrigation degrade and deplete the soil of its productive capacity
9	National Economic Empowerment and Development Strategy, 1999 (Ozoani, 2019)	a. To offer farmers improved irrigation, machinery and crop varieties to boost agricultural productivity and tackle poverty	 a. Lack of coordination between and among the stakeholders at different level of government b. lack of fund to pursue specific programme to an expected end
10	National Agricultural Land Development Authority, 1992 (Ozoani, 2019).	 a. To give strategic public support for land development. b. To assist and promote better uses of Nigeria's rural land and their resources, 4 c. To raise the living standard of rural people towards achieving food security. 	 a. The programme collapses due to usurp land that belonged to poor people by highly places officers. b. Lack of adequate data for decision making.
11	National, Special Programme on Food Security, 2002 (Iwuchukwu & Igbokwe, 2012).	 a. To increase food production and eliminate rural poverty b. To assist farmers, increase their output, productivity and income; strengthening the effectiveness of 	 a. Inability of majority of the beneficiaries to repay the loan on time b. Complexity in integrating technology into existing production system.

		research and extension service training and To educate farmers on farm management on effective resource utilisation. I. To support governments efforts in the promotion of simple technologies for self-sufficiency.
12	Root and Tuber Expansion Programme, 2003 (Iwuchukwu & Igbokwe, 2012).	 a. To address the problem of food production and rural poverty b. To achieve economic growth and improve access of the poor to social services c. To achieve food security and stimulate demand for cheaper root and tuber staple food. a. Inadequate virile technical extension services. b. The programme suffered short duration and continuity.
13	Nigeria Agro-Dealer Support Project, 2008 (Grow Africa, 2016).	 a. To compile comprehensive directory of agro-dealers to assess where and how product flow could operate through the private sector b. Poor record keeping and reconciliation process c. Rigorous registration and selection process which turned off input suppliers and agro- dealer capacity c. To strengthen marketing skills and provide technical services such as field demonstrations, soil testing and best agricultural practices to farmers.
14	Comprehensive Africa Agriculture Development Programme, 2009 (Ibietan, 2011)	 a. To strengthen agriculture through regulatory transformation b. To creating robust funding through budgetary provisions, subsidies and concessionary loans with single-digit interest. a. Non interaction between and among stakeholders b. Conflicting role between different programmes and projects c. inadequate monitoring and evaluation of programme

December 2021

		 c. To stabilize market prices and maximizing agricultural revenue in key enterprises. d. To take effective care and management of water and aquatic resources 	
15	Agriculture Transformation Agenda, 2011 (Grow Africa, 2016)	 a. To industrialize the agricultural sector to become more productive, efficient and competitive through a government-supported but private sector-led agricultural transformation b. To reduce reliance on food imports by adding 20 million metric tonnes to food supply c. To expand value addition to locally produced agricultural production d. Create 3.5 million jobs along agricultural value chains 	 a. Lack of fund to pursue the programme to an expected end b. Lack of continuity in commitment to policy c. Political opposition during implementation
16	Nigeria Incentive Based Risk Sharing System for Agricultural Lending, 2011. (FMARD, 2011)	 a. To generate a \$350 million risk-sharing facility to reduce the risk of lending by banks to farmers and agri- businesses b. To unlock access to input suppliers, agro- processors and product marketers c. To introduce a holistic approach that addresses both finance and agricultural value chains 	a. Strict loan policy which bar smallholder farmers loan accessibility
17	Growth Enhancement Scheme, 2012(Grow Africa, 2016)	a. To target individual smallholder farmers through a smart subsidy, using a high-tech delivery mechanism	 a. Compromise during implementation which defeat policy purposes b. Political insensitivity to policy demands

		 b. To increase farmer access to and use of improved seed and fertilizer by smallholder farmers at 50 per cent subsidy c. To break the cycle of inefficient and ineffective fertilizer and seeds support delivery to the targeted beneficiaries. 	c. Corruption and lack of adequate data for decision making.
18	Fund for Agricultural Finance in Nigeria, 2013 (Grow Africa, 2016)	 a. To generate inclusive growth in agriculture and to increase commercial capital available for agriculture. b. To tailor, long-term capital to small and medium enterprises across the agriculture value chain, 	 a. Rigorous registration and selection process which turned off most farmers b. Inadequate monitoring and evaluation of the programme

Source: Author's compilation.

Figure 4: Implementation Challenges of Agricultural Subsidy Regimes around the World

